

Format for ANSWERING REVIEWERS



September 6, 2013

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 4402-review.doc).

Title: Sonographic markers for early diagnosis of fetal malformations

Author: Maria Daniela Renna, Paola Pisani, Francesco Conversano, Emanuele Perrone, Ernesto Casciaro, Gian Carlo Di Renzo, Marco Di Paola, Antonio Perrone, Sergio Casciaro

Name of Journal: *World Journal of Radiology*

ESPS Manuscript NO: 4402

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

- (1) The adult heart in Fig3 has been replaced by a fetal heart;
- (2) The description of Ebstein's defect has been improved;
- (3) A figure showing the 4 routine axial views of heart and great vessels has been added (Fig4);
- (4) The three main types of cardiovascular malformations in which prenatal diagnosis has been shown to be beneficial were specified
- (5) The inaccuracy about the treatment of simple CHD has been corrected
- (6) Labeling and captures of several figures were improved
- (7) An online supplement containing several more figures and cartoons to aid in understanding each condition that has been described in the text was added

3 References and typesetting were corrected

4 Table 1 with abbreviations has been added

Thank you again for publishing our manuscript in the *World Journal of Radiology*.

Sincerely yours,

A handwritten signature in black ink, appearing to read 'Sergio Casciaro'.

Sergio Casciaro, PhD Eng,
Institute of Clinical Physiology
National Council of Research (IFC-CNR)
c/o Campus Universitario Ecotekne
via per Monteroni, 73100 Lecce, Italy
E-mail: sergio.casciaro@cnr.it